Author's response to reviews

Title: Mycoplasma hominis necrotizing pleuropneumonia in an immunocompetent adolescent

Authors:

Andres Pascual (Andres.Pascual@chuv.ch)
Marie-Helene Perez (Marie-Helene.Perez@chuv.ch)
Katia Jaton (Katia.Jaton@chuv.ch)
Gaudenz Hafen (Gaudenz.Hafen@chuv.ch)
Stefano Di Bernardo (Stefano.Dibernardo@chuv.ch)
Jacques Cotting (Jacques.Cotting@chuv.ch)
Gilbert Greub (Gilbert.Greub@chuv.ch)
Bernard Vaudaux (Bernard.Vaudaux@chuv.ch)

Version: 3 Date: 8 August 2010

Author's response to reviews: see over
Re: Revised manuscript BMC ID 2010 # 1985328639296258 entitled "MYCOPLASMA HOMINIS NECROTIZING PLEUROPNEUMONIA IN AN PREVIOUSLY HEALTHY ADOLESCENT." by Pascual A. et al.

Dear Dr Alam,

We would like to thank you for the careful and constructive review of our manuscript. The Reviewers’ comments have been carefully addressed and point-by-point responses are provided in the original sequential order. The manuscript revised along the lines suggested by the Reviewers (all the modifications are highlighted) is enclosed.

We hope that the issues raised by the Reviewers have all been appropriately addressed and would like to thank you for considering our manuscript for publication in BMC Infectious Diseases.

Sincerely,

Andres Pascual, MD

Pediatric Department and Infectious Diseases Service, Centre Hospitalier Universitaire Vaudois (CHUV) and University of Lausanne; Rue de Bugnon 46, 1011, Lausanne, SWITZERLAND
Phone + 41 79 556 48 01 Fax +41 21 314 3759
E-mail: Andres.Pascual@chuv.ch

Enclosure: Revised manuscript, modifications highlighted